Author's response to reviews

Title: Behavioral and biochemical effects of pharmacopuncture (ST 36 and ST 25) in obese rats

Authors:

Mariana C Pontes (riavet@yahoo.com.br)
Lilian C Heck (lilianheck@gmail.com)
Janice C Coelho (janice.coelho@ufrgs.br)

Version: 3 Date: 11 February 2015

Author's response to reviews: see over
11th February, 2015

Dr. James Ellis D. Prozenko
Journal Editorial Office
BMC Complementary and Alternative Medicine

REF: MS: 1183779667149875

Dear Sir,

We are resubmitting the manuscript “Behavioral and biochemical effects of pharmacopuncture (ST 36 and ST 25) in obese rats”. All changes were introduced in the text.

Please find below a “point to point” list of changes.

I look forward to hearing from you in a near future.

Sincerely yours,

Mariana Chiste Pontes
Department of Biochemistry
Federal University of Rio Grande do Sul
Rua Ramiro Barcelos, 2600 anexo
90035-003 – Porto Alegre – RS
Brazil
e-mail: riavet@yahoo.com.br
phone: +55 51 33085550
fax: + 55 51 33085535

Point to point list of changes:

1) We included, in Methods (page 3, line 112) the pharmacopuncture details